9448007, ja, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2022GL099563 by University Of Southampton, Wiley Online Library on [3/1/0/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

In situ Particle Measurements Deemphasize the Role of Size in Governing the Sinking Velocity of Marine Particles

J. R. Williams¹, S. L. C. Giering¹

¹National Oceanography Centre, Southampton, UK

Corresponding author: Jack Williams (jrw1n17@soton.ac.uk)

Key Points:

- Ex-situ studies typically find size to be a strong predictor of sinking velocity; strong correlations are rarely observed in situ however.
- Increased homogeneity of other particle characteristics ex situ (e.g. density, composition) are responsible for this discrepancy.
- Results suggest importance of these other factors when predicting particle sinking velocities and fluxes using size-scaling relationships.

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process, which may lead to differences between this version and the Version of Record. Please cite this article as doi: 10.1029/2022GL099563.

Abstract

Sinking particles are important in delivering carbon to the deep ocean where it may be stored out of contact with the atmosphere. Whilst particle sinking velocities are known to be influenced by a multitude of factors, size-based parameterisations remain common in biogeochemical models and in the methods used to determine particulate fluxes from autonomous platforms. Here we carried out an extensive literature review (62 datasets) into the size-sinking velocity relationship, and find the relationship is much weaker for studies examining particles in situ (median $R^2 = 0.03$) compared with ex situ studies (median $R^2 = 0.35$). This discrepancy may be because particles examined in the laboratory have more uniform properties than those studied in situ. Our review highlights the shortcomings of using a simple relationship between size and sinking velocity to calculate sinking particulate fluxes in the ocean; considering additional particle characteristics will enable more accurate calculations of particulate fluxes.

Plain Language Summary

Sinking particles are important in delivering carbon to the deep ocean where it may be stored out of contact with the atmosphere. Sinking particle speeds are an important control on sinking particle fluxes, and are known to be influenced by many factors. However, methods used to study particle fluxes from particle image datasets commonly place size as an important determinant of sinking speed. Here we carried out an extensive literature review into the size-sinking speed relationship, and find the relationship is much weaker for studies measuring particle sinking speeds in the ocean (in situ studies) compared with studies measuring sinking speeds in the laboratory (ex situ studies). This may be because other particle characteristics that can also influence particle sinking speeds are more constant in ex situ studies, allowing size to exert a stronger control on particle sinking speed. These results have implications for the use of novel in-situ imaging methods to calculate sinking particle fluxes, highlighting that it is important to incorporate information about particle characteristics as well as size when using these methods.

1 Introduction

In the ocean, the production, transfer to depth, and remineralization of organic particles provide a major pathway for the export of carbon from the ocean's surface to the ocean interior (Volk &

Hoffert, 1985). Collectively termed the Biological Carbon Pump (BCP), these processes act to maintain atmospheric CO₂ approximately 200 ppm lower than they would otherwise be (Maier-Reimer et al., 1996; Parekh et al., 2006). Although several processes contribute to the BCP, the gravitational settling of organic particles are thought to result in ~1000 Pg of ocean carbon storage (Boyd et al., 2019), up to 90% of the carbon sequestered by the BCP (Boyd et al., 2019; Buesseler et al., 2020; Sarmiento & Gruber, 2006).

As particulate organic carbon (POC) sinks, proportions of this downward flux are reworked by metazoans such as zooplankton, and eventually remineralised back into CO₂, through both microbial and zooplankton respiration (Giering et al., 2014; Steinberg *et al.*, 2008; van der Jagt *et al.*, 2020). As a result of this particle remineralisation and reworking, sinking POC fluxes are observed decrease with depth. The rate of flux attenuation (and hence the proportion of sinking carbon reaching the deep ocean) is determined by the balance between particle sinking velocities and remineralisation rates (Bach et al., 2019; Marsay et al., 2015). Since particle sinking velocities determine the length of time in which a particle is exposed to metazoan and microbial remineralisation, sinking velocity is a crucial determinant in the degree of attenuation of POC fluxes and BCP efficiency (Laurenceau-Cornec et al., 2015) (Fig. 1).

In recent years, the use of in situ optical methods has emerged as an important tool in the study of the BCP (Giering et al., 2020a). Increasingly able to be deployed autonomously (Lombard et al., 2019; Picheral et al., 2022), these methods can provide far greater spatiotemporal resolution and coverage than traditional ship-based sampling methods (Giering et al., 2020a; Lombard et al., 2019). Given also the considerable effort that has focused on improving the utility of in situ imaging devices, in situ cameras are now capable of providing quantitative particle information on particles from $1 - 10,000 \mu m$ in diameter (Lombard *et al.*, 2019). Using the particle size distributions obtained by in situ imaging methods, particle fluxes within a given size class can be calculated if sinking velocities of particles within the size class can also be estimated (McDonnell & Buesseler, 2010; 2012), or, more commonly, through directly relating particle size to flux via an empirical relationship (Guidi *et al.*, 2008; Iversen *et al.*, 2010). A robust understanding of the factors that govern particle sinking rate is therefore

crucial in the implementation of these cutting-edge methods for estimating particulate fluxes and studying the BCP.



Figure 1: Schematic illustrating factors posed to influence particle sinking velocity, their relation to size (red) or density (green) as described in Stokes's law, and the effect of fast and slow sinking particles on particulate flux attenuation rates and particles fluxes reaching depth. Illustrative flux attenuation curves are shown for fast-sinking (left) and slow-sinking (right) particles. Fast sinking particles experience slower rates of flux attenuation due to decreased duration of exposure to remineralisation whilst they sink.

Whilst the use of particle size spectra provide the means to calculate particulate fluxes with unprecedented resolution, the prevailing methods used to empirically relate size to fluxes

assume that mass and sinking velocity as functions of particle size remain constant (Guidi et al., 2008; Iversen et al., 2010). Further, relying on a single relationship to describe sinking velocity in terms of particle size implies that larger particles should always sink faster than smaller ones (McDonnell & Busesseler, 2010). Although several studies have observed size to exert a strong control on sinking velocities (Alldredge & Gottschalk, 1988; Iversen et al., 2010), considering size as a key predictor of sinking velocity assumes relative constancy of other particle properties such as particle composition, porosity and biomineral content. In recent years, empirical studies have indicated this assumption of constancy can not be applied in situ (Iversen & Ploug, 2010; Laurenceau-Cornec et al., 2020; Ploug et al., 2008) and that size alone is often a poor predictor of sinking velocity (Diercks & Asper, 1997; Iversen & Lampitt, 2020). Nevertheless, size-based methods remain a commonly used approach to estimate fluxes from in situ image data (Cram *et al.*, 2022; Fender *et al.*, 2019; Guidi *et al.*, 2016; Kiko *et al.*, 2017; 2020; Ramondenc *et al.*, 2016), and size-based parameterisations of vertical carbon fluxes remain common in biogeochemical models (Aumount *et al.*, 2015; Kriest & Oschlies, 2008; Leung *et al.*, 2021; Swart *et al.*, 2019; Yool *et al.*, 2021).

Here we first outline the theoretical basis underpinning commonly used size-based approaches. We then highlight the lack of evidence suggesting size can represent a strong predictor of marine particle sinking velocities in situ, and assess the reasons for differences between studies. We recommend avenues for further study that will facilitate improved mechanistic understanding of particle sinking velocities and broaden the applicability of in situ image-based estimations of particle flux.

2.1 The theoretical basis for size-based methods

In recent decades, derivations such as Stokes's law have been widely used to estimate particle sinking velocity (Laurenceau-Cornec et al., 2020). Assuming that particle drag coefficients can be calculated as a simple function of Reynolds number for low Reynolds numbers (in laminar flow conditions), and balancing drag and gravitational forces on a particle, these derivations pose size to be a key determinant of sinking velocity. Stokes's Law says that

$$w = \left(\rho_p - \rho_f\right) \frac{gD^2}{18\mu} \tag{2},$$

where *w* is the sinking velocity of a sphere (m s⁻¹), ρ_p and ρ_f are the sphere and fluid densities (kg m⁻³), *g* is the acceleration due to gravity (9.81 m s⁻²), *D* is the sphere diameter (m) ,and μ is the fluid dynamic viscosity in kg m⁻¹s⁻¹. Power law functions based on Stokes's Law have been used to relate particle size to sinking velocity for decades (Alldredge & Gotschalk, 1988; Smayda, 1970) and more recently to parameterise modelled particle sinking velocities (Aumont *et al.*, 2015; DeVries *et al.*, 2014; Kriest & Evans, 1999; Leung *et al.*, 2021). Other biogeochemical models simply incorporate size through discrete size classes, with a large, fastsinking fraction, and a small, slow-sinking fraction (Aumont *et al.*, 2015; Swart *et al.*, 2019; Yool *et al.*, 2021). The lack of mechanistic understanding as to how well size constrains sinking velocity has resulted in a variety of size-sinking relationships in earth system models, which yield up to order of magnitude differences in sinking velocity for particles of the same size, and introduce uncertainty into flux prediction and biogeochemical models (Cael *et al.*, 2021; Niemeyer *et al.*, 2019). Since plankton models additionally suggest a decrease in cell size with warming, constraining sinking velocities in an accurate mechanistic fashion is of importance for the accurate modelling of climate change projections (Cael *et al.*, 2021; Finkel *et al.*, 2010).

Power law functions have more recently also been used to directly estimate particulate fluxes from particle size distributions. Since both sinking velocity (*w*) and particle mass (*m*) and hence flux for a given particle *i* (*F_i*) can be expressed as power law functions of the form ($y = ax^b$), their product can be expressed in the same form:

$$F_i = wm = AD^B \tag{3},$$

where *D* is particle diameter, and *A* and *B* are constants (Guidi *et al.*, 2008). If *A* and *B* are known, size spectra can be used to calculate total mass fluxes, *F*. *A* and *B* may be estimated through a minimisation procedure (Cram *et al.*, 2022; Fender *et al.*, 2019; Guidi et al., 2008; Iversen et al., 2010; Nowald et al., 2015) if alternative measurements of particulate fluxes can be made, and assuming that mass and particle size as a function of depth are constant for all depths (Iversen et al., 2010). Alternatively, when additional flux measurements have not been made (such as on autonomous deployments on moorings, gliders, or floats), prior studies can be used to estimate global values for *A* and *B* (Guidi *et al.*, 2008; 2016; Kiko *et al.*, 2020; Ramondenc et

al., 2016). The above approaches assume that particle mass and sinking velocity as functions of size are constant with depth and, in the latter instance, universally constant; hence both methods represent size as a strong control of sinking velocity.

2.2 Empirical evidence on the size-sinking velocity relationship

To direct our discussion in the most constructive fashion towards particle characteristics commonly discussed in the literature, we focused our analysis on four of the most commonly studied characteristics. It should however be noted that the frequency with which characteristics are discussed in the literature does not necessarily indicate that they are the most important four drivers of sinking velocity. To identify these characteristics, we carried out a literature search into studies measuring particle sinking velocity and associated particle characteristics using both in situ and ex situ methods, and commonly used keyterms to describe marine particle characteristics ("Size", "Ballast", "Morphology", "Composition", "Type", "Shape", "Compactness", "Fractal" [Dimension]). Restricting results to within Earth and Planetary Sciences, we searched for abstracts, titles, and keywords containing the words "Particle" and "Sinking" and "Velocity" as well as a given particle characteristic. Size returned the greatest number of studies (79), followed by parameters relating to chemical and taxonomic composition (Composition: 37; Type: 22; Ballast: 18). Searches relating to other morphological properties typically returned the fewest results ("Shape": 17; "Fractal" [Dimension]: 5; "Compactness": 1; "Permeability": 1; "Morphology": 1).

Using the four most commonly studied particle attributes from our literature search (size, particle type, ballast, and shape), we identified 62 dataset from 38 studies and examined the degree of correlation between sinking velocity and each of the above attributes (Fig. 2). A full description of methods is provided in the *Methods* section. Briefly, for particle type, ballast, and shape, R² (proportion of variance in sinking velocity explained by size) were calculated from linear regressions for continuous variables, or from analyses of variance (ANOVAs) for categorical variables. To assess the degree of variation in sinking velocity explained by particle size in each study, a power law function was fitted to the data, and R² of this power law function calculated using a linear regression on the log-log plot of size against sinking velocity. A power law function was chosen over a linear relationship since sinking velocity is thought to scale with particle diameter according to a power law function according to Stokes's Law and empirically

modified versions incorporating porosity (Guidi et al., 2008; Laurenceau-Cornec et al., 2020; Xiang et al., 2022).

Despite particle size having received the most interest, our review suggests the dependency of sinking velocity on size is not well constrained (Fig. 2). Size explains between 0-88% in the variation in particle sinking velocity (as determined by the coefficient of determination (R^2)) with a median value of 31%. The strongest correlation between size and sinking velocity was observed for intact salp faecal pellets from the Southern Ocean (Iversen et al., 2017). However, in 26% of the datasets, particle size was observed to be a poor predictor of sinking velocity, explaining less than 10% of variation in particle sinking velocity (Fig. 2). Particle size did not appear to be a stronger predictor of sinking velocity than particle type or particle ballast content (Wilcoxon rank tests, p > 0.8). The median percentage of variance in sinking velocity explained by particle type and ballast content were 26% and 30%, respectively. It is noteworthy however that only four ex situ datasets examined the influence of particle type (i.e. differences in both taxonomic composition, i.e. aggregates made of different phytoplankton species, or particle shape, e.g. faecal pellets vs aggregates) or ballast content. Likewise for particle shape, only one ex situ (Laurenceau-Cornec et al., 2015) and one in situ (Iversen & Lampitt, 2020) study directly measured a particle shape characteristic (aspect ratio) and sinking velocity, with neither of these studies finding sinking velocity to be explained by particle shape alone.

For the datasets focussing on particle size as a predictor, we found strong differences between measurements made in situ and ex situ. R^2 values were significantly higher for ex situ studies than in situ studies (Wilcoxon rank test, p < 0.01), suggesting that the strength of the size-sinking velocity relationship may be influenced by whether measurements are made in or ex situ. While weak correlations between size and sinking velocity were observed in both situ and ex situ datasets, strong dependencies of sinking velocity on particle size were only observed ex situ. However, when combining all ex situ datasets, a clear lack of a 'global' size-to-sinking velocity relationship becomes apparent (n = 4138, p = <0.001, R² = 0.193);Fig. 3a; though note that these studies used different method, which may forego a direct comparison (Giering et al. 2020b). For in situ datsets (n = 12), size explained less than 30% of variability in sinking velocity in all but two studies which respectively examined flocs from meltwater discharge plumes and resuspended near-bottom sediment. For in situ particles, the median percentage of variance in sinking velocity explained by particle size was 3%, contrasting with 35% for particles measured ex situ. Overall these findings suggest that the strong relationships observed ex situ between individual particle characteristics and sinking velocity rarely hold true in situ. As such, the methodological biases outlined below should be taken into consideration before extrapolating relationships observed in ex situ studies to natural marine particles in situ.



Figure 2: Boxplot comparing proportion of variance in sinking velocity explained by particle characteristics. Coefficients of determination (R²) from linear models and analyses of variance (ANOVAs) performed between particle characteristics (size, type, ballast content, shape) and sinking velocities directly measured in previous studies (see text, Supplementary Table S1). Colours of boxplots indicate whether sinking velocity measurements were made in situ (dark grey) or ex situ (light grey), and how particles were generated (in situ: natural particles observed in situ or measured immediately ex situ without prior incubation; ex situ: cultured or incubated

ex situ prior to measurement. Shapes of the points indicate method used to measure sinking velocity.

2.3 Ex situ vs in situ: methodological compromises

Methodological differences between in situ and ex situ studies can explain both the predominance of ex situ studies examining the size-sinking velocity relationship, and why size is seen to more tightly constrain sinking velocities ex situ than in situ.

In situ methods hold the major advantage of observing particles in their natural environment. Any measurements made are therefore acquired without the need for handling particles, decreasing (but not eliminating (Briggs et al., 2011; Cetinić et al., 2012)) the potential alteration and disturbance to particle properties (Giering et al., 2020a; Iversen & Lampitt, 2020), and thus maximising the realism of any interactions between sinking velocity and particle characteristics. However, a major drawback of in situ optical devices is that these methods lack the capability to provide direct information on a number of particle characteristics, such as particle density and composition (Giering et al., 2020b). These methods must hence rely on additional data or assumptions to estimate particle sinking velocities and calculate particulate fluxes. Given these uncertainties, the expensive nature of in situ camera systems, and a lack of standardization in analysis routines for in situ image datasets (Giering et al., 2020b), in situ studies into particle sinking velocities remain sparse compared with more traditional ex situ methods. In summary, in situ studies lack the capacity to study particle characteristics which may be measured ex situ, but maximise realism (Fig. 3b).

Most studies into factors constraining sinking velocity involve incubating particles ex situ prior to or during measurements. By examining particles in a laboratory, detailed measurements of a wide number of particle characteristics can be made, such as chemical and taxonomic composition, removing the need for estimates of these parameters (Mantovanelli & Ridd, 2006). In addition, studies where particles are generated ex situ also allow for manipulation of particles characteristics, to test for effects of specific particle characteristics on sinking velocities (Giering *et al.*, 2020a). However, ex situ particles are likely not reflective of in situ particle dynamics,

partly owing to ex situ particles being more homogenous. Three aspects contribute to this discrepancy: (1) Homogeneity in the 'source' particle pool; (2) homogenization of particles during particle collection; (3) homogenization of particles during the incubation for measuring sinking velocities.

Firstly, particles for ex situ incubations are often sourced from "artificial", laboratory-produced particles, e.g. incubating homogenous particle pools such as phytoplankton cultures in roller tanks, whilst in the natural environment a heterogenous pool of particles of varied age, composition, density, structure, and porosity exists (Alldredge, 1998; Alldredge & Gotschalk, 1988; Iversen & Lampitt, 2020). The unnatural homogeneity of ex situ particles sourced in this way reduces the variability in sinking velocity introduced by factors other than size, thus allowing size to exert a dominant control over sinking velocity.

An alternative approach to sourcing particles for incubations involves the collection of natural marine particles. Whilst this approach allows for collection of a more diverse particle pool, highly fragile marine aggregates are susceptible to damage, alteration, and compaction or disaggregation during sampling for ex situ incubations (Alldredge & Gotschalk, 1988; Alldredge & Silver, 1988; Giering et al., 2020a; Iversen & Lampitt, 2020; Kajihara, 1971; Takeuchi et al., 2019). Particles measured in the laboratory are able to withstand higher turbulences than that observed in the ocean, (Alldredge et al., 1990; Riebesell, 1992), and also typically exhibit increased sinking velocities compared with measurements made in situ (Fig. 3a) (Alldredge & Gottschalk, 1988; Shanks & Trent, 1980). These observations indicate that, despite the more heterogenous nature of naturally occurring particles compared to cultured particles, sampling in this way has a tendency to alter particles (and/or particle populations) such that they are no longer fully representative of marine particles in situ.

Lastly, particles incubated ex situ are exposed to a far more limited set of processes influencing their formation and composition. In situ, a number of biological and physical processes contribute to the aggregation of particles. For physical processes, mechanisms such as Brownian motion, differential settling (in which which faster sinking particles scavenge slower sinking or suspended particles upon collision), and turbulent shear (McCave, 1984) influence aggreate formation, with the importance of these processes varying depending on particle size (Jackson,

1994; McCave, 1984; Takeuchi et al., 2019). By contrast, in roller tanks turbulent shear is negligible in aggregate formation once the initial spin-up period is over (Engel et al., 2009; Laurenceau-Cornec et al., 2015), whilst in Couette chambers the influence of shear is amplified relative to in situ (Jackson, 2015; Lick et al., 1993). Ex situ incubations also typically lack the diversity of biologically-mediated processes that aggregate or disaggregate particles. In situ, aggregation processes in situ can include compaction into faecal pellets, the accretion of particles onto mucous houses and other exuded exopolymers (Hamner et al., 1975; Hansen et al., 1996; Kiørboe, 2001), and aggregation due to organisms' feeding currents (Fukuda & Koide, 2000). Particle disaggregation as a result of zooplankton feeding can also occur in situ, a process which may not be included in ex situ incubations (Dilling & Alldredge, 2000; Iversen & Poulsen, 2007). Given that in situ studies cannot replicate the diversity of in situ processes involved in forming and transforming particles, it is unsurprising that ex situ particle pools are more homogenous that those in situ, and hence exhibit stronger size-scaling relationships. Overall, ex situ studies favour the ability to measure and examine particle dynamics in detail, whilst sacrificing realism (Fig. 3b).



Figure 3: (a) Particle size (equivalent spherical diameter, ESD) against sinking velocity for all particles in this study, from both in situ (blue) and ex situ (orange) studies.; (b) Relative

advantages of in situ and ex situ methods (blue and orange respectively) in terms of ability to measure particle characteristics and particle realism, when investigating the relationship between particle sinking velocity and particle characteristics. Position of each method relates to an assigned "Particle Realism score" and "Measurement Capability score" as described in *4.3 Methodological Comparisons*; explanations of assertions used for scoring these methods available in Supplementary Table S2.

3. An outlook for the use of in situ methods

This review highlights the discrepancy in the extent to which size controls sinking velocity between in situ and ex situ studies (Alldredge & Gotschalk, 1988), and that - despite this discrepancy- size-based methods remain common for estimating fluxes from in situ data or representing marine particle particles in models.

While there is clear merit in using simple size-to-sinking velocity relationships for autonomous methods (e.g. Fender et al., 2019; Guidi et al., 2008; Iversen et al., 2010), these approaches provide limited mechanistic understanding into the size-flux relationship, limiting the certainty with which relationships can be spatiotemporally extrapolated. As suggested by McDonnell & Buesseler (2010), taking into account particle types will increase the range of spatiotemporal scales over which size-scaling relationships can be applied. Recognising individual size-scaling relationships for varying particle types will enable more accurate sinking velocity and flux estimates for each particle type. In turn, considering the weighted contribution of each particle type will maintain the accuracy of optical methods even under varied ecological and biogeochemical settings, when contributions from each particle type may vary. Alternatively, the inclusion of additional particle characteristics, such as compactness or bulk particle composition, into a unified equation (e.g. Giering et al. 2020b) may provide more accurate predictions of particle sinking velocities. Some information of these characteristics can be obtained from optical measurements, such as porosity (Bach et al., 2019), bulk density (Neukermans et al., 2012; Hurley et al., 2016), and bulk particle composition (inorganic/organic ratios, Twardowski et al., 2001; Loisel et al., 2007).

Yet, at present, the uncertainties associated with these proxies are large or unconstrained. Simultaneous measurements of particle type (or characteristics), size and sinking velocity will

enable the development of these methods, and is likely to be expedited by advances in machine learning (Giering *et al.*, 2020a; Iversen & Lampitt, 2020). Moving away from purely size-based velocity and flux relationships to incorporate these additional particle properties will not only facilitate improved mechanistic understanding of particle sinking and the BCP, but also promote increased spatio-temporal resolution of methods used to the study the BCP, through the use of autonomous platforms and in biogeochemical models.

4 Methods

4.1 Data compilation

We compiled observations of particle sinking velocity and associated particle characteristics from 62 datasets from 38 studies (see Supplementary Table S1). These data had previously been compiled by Cael et al. (2021) and Laurenceau-Cornec et al. (2015, 2020); all original datasets were validated and, if needed, redigitized using Plot Digitizer

(https://automeris.io/WebPlotDigitizer/). Studies not relating to marine particles were excluded from this analysis. In the small number of cases where particle size and sinking velocity data had been fitted to a power law function in original studies (n = 10), published R² values in the literature were used. Data were assigned to "in situ" and "ex situ" groups for measurement type, based on the method used to measure sinking velocity in each study. The particle types examined in each study were assigned to one of nine particle types (e.g., natural aggregates, mixed diatom culture; for full list Fig. 2, Supplementary Table S1), with method used to measure particle sinking velocities also described through one of nine groups (e.g., Scuba photography, Vertical flow system; for full list see legend of Fig. 2, Supplementary Table S1).

4.2 Sinking velocity/Particle characteristic analyses

To assess the variability in sinking velocity explained by a particle size in each study, a power law function (in form $w = Ad^B$, where w is the sinking velocity, d the diameter, and A and B are scaling coefficients) was fitted to the data.

For particle type, ballast, and shape, R^2 were recorded either from performing linear regressions or analyses of variance (ANOVAs), depending on whether the particle characteristic was described in terms of continuous or categorical data. For example, in some studies particle type

was analysed as a categorical variable with discrete groups such as *S. costatum* or *E. huxleyi* aggregates, and sinking velocity was compared between these groups by means of an ANOVA. In another study, particle type was expressed as a percentage of aggregate composition of one diatom morphotype (Laurenceau-Cornec *et al.*, 2015). In this case, a linear regression was performed between percentage of total composition and particle sinking velocity. Lastly, having failed both Levene's and Shapiro Wilk tests, a Wilcoxon rank sum test with continuity correction was performed to assess whether R^2 coefficients differed significantly between in situ and ex situ studies.

4.3 Methodological comparison

To represent the advantages and disadvantages of in situ and ex situ methods for sinking velocity measurement, methods were ranked in terms of their ability to measure particle characteristics, and in terms of particle realism. Although these assertions are subjective rankings, a scoring system was devised to standardise rankings and criteria by which methods were judged. For measurement capability score, particle characteristics (Size, Ballast, Taxonomic composition/Particle type, Chemical composition, Shape, Dry weight, Porosity, Fractal dimension, Density, and Sinking velocity) were assigned a score from 0 to 4, describing the comprehensiveness with which a particle characteristic could be studied with a given method (0 lowest, 4 highest; see Supplementary Table S2). Measurement capability scores of individual characteristics were summed to give an overall score. Where a range of measurement score was given for a particle characteristic, the mean value was used when summing scores to calculate (e.g. 2-3 scored as 2.5).

For the particle realism score, each method was assigned a score from 0 to 4, based on the extent to which the particles measured had been influenced by sampling and measurement procedures, i.e. the extent to which particle communities measured could be expected to reflect natural marine particle communities in situ. A brief explanation for assigned scores and evidence supporting these assertions are outlined in Supplementary Table S2).

Acknowledgments

The authors thank Emmanuel Laurenceau-Cornec, Morten Hvitfeldt Iversen, and B. B. Cael for their assistance with data acquisition; thanks goes also to authors of the original studies analyzed here for their efforts in collecting the data and providing it for our analysis. We thank also Mark Moore, the anonymous reviewers, and editors for their thoughtful suggestions and comments on this manuscript. JRW was supported by the ANTICS project, funded by a European Research Council (ERC) Starting Grant (EC-950212). The authors declare no competing financial interests.

Open Research

All data used in this work were accessed through previously published studies and are provided in Supplementary Data Set DS01. Supplementary Table S1 provides the data source details, and all studies are cited within our references. Supplementary Table S2 provides assertions underpinning schematic in Figure 3b and references underpinning these assertions.

References

Alldredge, A. (1998). The carbon, nitrogen and mass content of marine snow as a function of aggregate size. *Deep Sea Research Part I: Oceanographic Research Papers*, *45*(4-5), 529-541. https://doi.org/10.1016/S0967-0637(97)00048-4

Alldredge, A. L., & Gotschalk, C. (1988). In situ settling behavior of marine snow'. *Limnology* and Oceanography, 33(3), 339–374. https://doi.org/https://doi.org/10.4319/lo.1988.33.3.0339

Alldredge, A. L., & Gotschalk, C. C. (1989). Direct observations of the mass flocculation of diatom blooms: characteristics, settling velocities and formation of diatom aggregates. *Deep Sea Research Part A. Oceanographic Research Papers*, *36*(2), 159–171. https://doi.org/10.1016/0198-0149(89)90131-3

Alldredge, A. L., & Silver, M. W. (1988). Characteristics, dynamics and significance of marine snow. *Progress in oceanography*, 20(1), 41-82. https://doi.org/10.1016/0079-6611(88)90053-5

Alldredge, A. L., Granata, T. C., Gotschalk, C. C., & Dickey, T. D. (1990). The physical strength of marine snow and its implications for particle disaggregation in the ocean. *Limnology and Oceanography*, *35*(7), 1415-1428. https://doi.org/10.4319/lo.1990.35.7.1415

Aumont, O., Éthé, C., Tagliabue, A., Bopp, L., & Gehlen, M. (2015). PISCES-v2: an ocean biogeochemical model for carbon and ecosystem studies. *Geoscientific Model Development*, 8(8), 2465-2513. https://doi.org/10.5194/gmd-8-2465-2015

Azetsu-Scott, K., & Johnson, B. D. (1992). Measuring physical characteristics of particles: a new method of simultaneous measurement for size, settling velocity and density of constituent matter. *Deep Sea Research Part A. Oceanographic Research Papers*, *39*(6), 1057–1066. https://doi.org/10.1016/0198-0149(92)90039-V

Bach, L. T., Stange, P., Taucher, J., Achterberg, E. P., Algueró-Muñiz, M., Horn, H., Esposito,
M., & Riebesell, U. (2019). The Influence of Plankton Community Structure on Sinking Velocity
and Remineralization Rate of Marine Aggregates. *Global Biogeochemical Cycles*, *33*(8), 971–
994. https://doi.org/10.1029/2019GB006256

Belcher, A., Iversen, M., Manno, C., Henson, S. A., Tarling, G. A., & Sanders, R. (2016). The role of particle associated microbes in remineralization of fecal pellets in the upper mesopelagic of the Scotia Sea, Antarctica. *Limnology and Oceanography*, *61*(3), 1049–1064. https://doi.org/10.1002/LNO.10269

Belcher, A., Manno, C., Ward, P., Henson, S., Sanders, R., & Tarling, G. (2016). Zooplankton faecal pellet transfer through the meso-and bathypelagic layers in the Southern Ocean in spring 2. *Biogeosciences*, *13*(17), 4927–4943. https://doi.org/10.5194/bg-2016-520

Boyd, P. W., Claustre, H., Levy, M., Siegel, D. A., & Weber, T. (2019). Multi-faceted particle pumps drive carbon sequestration in the ocean. *Nature*, *568*(7752), 327–335. https://doi.org/10.1038/s41586-019-1098-2

Briggs, N., Perry, M. J., Cetinić, I., Lee, C., D'Asaro, E., Gray, A. M., & Rehm, E. (2011). Highresolution observations of aggregate flux during a sub-polar North Atlantic spring bloom. *Deep-Sea Research Part I: Oceanographic Research Papers*, *58*(10), 1031–1039. https://doi.org/10.1016/j.dsr.2011.07.007

19448007, ja, Downloaded from https://agupubs.onlinelibrary.wikey.com/doi/10.1029/2022GL099563 by University Of Southampton, Wiley Online Library on [31/10/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/term: and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creat

Buesseler, K. O., Boyd, P. W., Black, E. E., & Siegel, D. A. (2020). Metrics that matter for assessing the ocean biological carbon pump. *Proceedings of the National Academy of Sciences*, *117*(18), 9679–9687. https://doi.org/10.1073/pnas.1918114117/-/DCSupplemental

Cael, B. B., Cavan, E. L., & Britten, G. L. (2021). Reconciling the Size-Dependence of Marine Particle Sinking Speed. *Geophysical Research Letters*, *48*(5), e2020GL091771. https://doi.org/10.1029/2020GL091771

Carder, K. L., Steward, R. G., & Betzer, P. R. (1982). In Situ Holographic Measurements of the Sizes and Settling Rates of Oceanic Particulates. *Journal of Geophysical Research*, 87(C8), 5681–5685. https://doi.org/10.1029/JC087iC08p05681

Cavan, E. L., Giering, S. L. C., Wolff, G. A., Trimmer, M., & Sanders, R. (2018). Alternative Particle Formation Pathways in the Eastern Tropical North Pacific's Biological Carbon Pump. *Journal of Geophysical Research: Biogeosciences*, *123*(7), 2198–2211. https://doi.org/10.1029/2018JG004392

Cetinić, I., Perry, M. J., Briggs, N. T., Kallin, E., D'asaro, E. A., Lee, C. M., Cetinić, C. :, Perry, M. J., Briggs, N. T., Kallin, E., D'asaro, E. A., & Lee, C. M. (2012). Particulate organic carbon and inherent optical properties during 2008 North Atlantic Bloom Experiment. *Journal of Geophysical Research*, *117*(C6), 6028. https://doi.org/10.1029/2011JC007771

Cram, J. A., Fuchsman, C. A., Duffy, M. E., Pretty, J. L., Lekanoff, R. M., Neibauer, J. A., ... & McDonnell, A. M. (2022). Slow particle remineralization, rather than suppressed disaggregation, drives efficient flux transfer through the Eastern Tropical North Pacific Oxygen Deficient Zone. *Global Biogeochemical Cycles*, *36*(1), e2021GB007080. https://doi.org/10.1029/2021GB007080

Deibel, D. (1990). Still-water sinking velocity of fecal material from the pelagic tunicate Dolioletta gegenbauri. *Marine Ecology Progress Series*, 62, 55–60.

DeVries, T., Liang, J. H., & Deutsch, C. (2014). A mechanistic particle flux model applied to the oceanic phosphorus cycle. *Biogeosciences*, *11*(19), 5381-5398. https://doi.org/10.5194/bg-11-5381-2014

Diercks, A.-R., & Asper, V. L. (1997). In situ settling speeds of marine snow aggregates below the mixed layer: Black Sea and Gulf of Mexico. *Deep Sea Research Part I: Oceanographic Research Papers*, 44(3), 385–398. https://doi.org/https://doi.org/10.1016/S0967-0637(96)00104-5

Dilling, L., & Alldredge, A. L. (2000). Fragmentation of marine snow by swimming macrozooplankton: A new process impacting carbon cycling in the sea. *Deep Sea Research Part I: Oceanographic Research Papers*, 47(7), 1227-1245. https://doi.org/10.1016/S0967-0637(99)00105-3

Engel, A., & Schartau, M. (1999). Influence of transparent exopolymer particles (TEP) on sinking velocity of Nitzschia closterium aggregates. *Marine Ecology Progress Series*, *182*, 69–76. https://doi.org/10.3354/MEPS182069

Engel, A., Szlosek, J., Abramson, L., Liu, Z., & Lee, C. (2009). Investigating the effect of ballasting by CaCO₃ in Emiliania huxleyi: I. Formation, settling velocities and physical properties of aggregates. *Deep Sea Research Part II: Topical Studies in Oceanography*, *56*(18), 1396–1407. https://doi.org/10.1016/J.DSR2.2008.11.027

Fender, C. K., Kelly, T. B., Guidi, L., Ohman, M. D., Smith, M. C., & Stukel, M. R. (2019). Investigating particle size-flux relationships and the biological pump across a range of plankton ecosystem states from coastal to oligotrophic. *Frontiers in Marine Science*, *6*, 603. https://doi.org/10.3389/fmars.2019.00603 Finkel, Z. V., Beardall, J., Flynn, K. J., Quigg, A., Rees, T. A. V., & Raven, J. A. (2010). Phytoplankton in a changing world: cell size and elemental stoichiometry. *Journal of plankton research*, *32*(1), 119-137. https://doi.org/10.1093/plankt/fbp098

Fowler, S. W., & Small, L. F. (1972). SINKING RATES OF EUPHAUSIID FECAL PELLETS. *Limnology and Oceanography*, *17*(2), 293–296. https://doi.org/10.4319/LO.1972.17.2.0293

Francis, S., & Passow, U. (2020). Transport of dispersed oil compounds to the seafloor by sinking phytoplankton aggregates: A modeling study. *Deep-Sea Research Part I: Oceanographic Research Papers*, *156*, 103192. https://doi.org/10.1016/j.dsr.2019.103192

Fukuda, H., & Koide, I. (2000). Feeding currents of particle-attached nanoflagellates a novel mechanism for aggregation of submicron particles. Marine Ecology Progress Series, 202, 101-112. doi:10.3354/meps202101

Giering, S. L. C., Cavan, E. L., Basedow, S. L., Briggs, N., Burd, A. B., Darroch, L. J., Guidi, L.,
Irisson, J. O., Iversen, M. H., Kiko, R., Lindsay, D., Marcolin, C. R., McDonnell, A. M. P.,
Möller, K. O., Passow, U., Thomalla, S., Trull, T. W., & Waite, A. M. (2020a). Sinking Organic
Particles in the Ocean—Flux Estimates From in situ Optical Devices. *Frontiers in Marine Science*, *6*, 834. https://doi.org/10.3389/fmars.2019.00834

Giering, S. L. C., Hosking, B., Briggs, N., & Iversen, M. H. (2020b). The Interpretation of Particle Size, Shape, and Carbon Flux of Marine Particle Images Is Strongly Affected by the Choice of Particle Detection Algorithm. *Frontiers in Marine Science*, *7*, 564. https://doi.org/10.3389/fmars.2020.00564

Giering, S. L. C., Sanders, R., Lampitt, R. S., Anderson, T. R., Tamburini, C., Boutrif, M., Zubkov, M. v., Marsay, C. M., Henson, S. A., Saw, K., Cook, K., & Mayor, D. J. (2014). Reconciliation of the carbon budget in the ocean's twilight zone. *Nature*, *507*(7493), 480–483. https://doi.org/10.1038/nature13123

19448007, ja, Downloaded from https://agupubs.onlinelibrary.wikey.com/doi/10.1029/2022GL099563 by University Of Southampton, Wiley Online Library on [31/10/2022]. See the Terms and Conditions (https://onlinelibrary.wiley.com/ conditions) on Wiley Online Library are governed by the applicable Creat

Guidi, L., Chaffron, S., Bittner, L., Eveillard, D., Larhlimi, A., Roux, S., ... & Gorsky, G. (2016). Plankton networks driving carbon export in the oligotrophic ocean. *Nature*, *532*(7600), 465-470. https://doi.org/10.1038/nature16942

Guidi, L., Jackson, G. A., Stemmann, L., Miquel, J. C., Picheral, M., & Gorsky, G. (2008). Relationship between particle size distribution and flux in the mesopelagic zone. *Deep-Sea Research Part I: Oceanographic Research Papers*, *55*(10), 1364–1374. https://doi.org/10.1016/j.dsr.2008.05.014

Hamner, W. M., Madin, L. P., Alldredge, A. L., Gilmer, R. W., & Hamner, P. P. (1975). Underwater observations of gelatinous zooplankton: sampling problems, feeding biology, and behavior 1. *Limnology and Oceanography*, *20*(6), 907-917. https://doi.org/10.4319/lo.1975.20.6.0907

Hansen, J. L., Kiørboe, T., & Alldredge, A. L. (1996). Marine snow derived from abandoned larvacean houses: sinking rates, particle content and mechanisms of aggregate formation. *Marine Ecology Progress Series*, *141*, 205-215. doi:10.3354/meps141205

Hawley, N. (1982). Settling Velocity Distribution of Natural Aggregates. *Journal of Geophysical Research*, 87(C12), 9489–9498. https://doi.org/10.1029/JC087iC12p09489

Hill, P. S., Syvitski, J. P., Cowan, E. A., & Powell, R. D. (1998). In situ observations of floc settling velocities in Glacier Bay, Alaska. *Marine Geology*, *145*(1–2), 85–94. https://doi.org/10.1016/S0025-3227(97)00109-6

Hurley, A. J., Hill, P. S., Milligan, T. G., & Law, B. A. (2016). Optical methods for estimating apparent density of sediment in suspension. *Methods in Oceanography*, *17*, 153-168. https://doi.org/10.1016/j.mio.2016.09.001 Iversen, M. H., & Lampitt, R. S. (2020). Size does not matter after all: No evidence for a sizesinking relationship for marine snow. *Progress in Oceanography*, *189*, 102445. https://doi.org/10.1016/j.pocean.2020.102445

Iversen, M. H., Nowald, N., Ploug, H., Jackson, G. A., & Fischer, G. (2010). High resolution profiles of vertical particulate organic matter export off Cape Blanc, Mauritania: Degradation processes and ballasting effects. *Deep-Sea Research Part I: Oceanographic Research Papers*, *57*(6), 771–784. https://doi.org/10.1016/j.dsr.2010.03.007

Iversen, M. H., Pakhomov, E. A., Hunt, B. P. V., van der Jagt, H., Wolf-Gladrow, D., & Klaas, C. (2017). Sinkers or floaters? Contribution from salp pellets to the export flux during a large bloom event in the Southern Ocean. *Deep Sea Research Part II: Topical Studies in Oceanography*, *138*, 116–125. https://doi.org/10.1016/J.DSR2.2016.12.004

Iversen, M. H., & Ploug, H. (2010). Ballast minerals and the sinking carbon flux in the ocean:
Carbon-specific respiration rates and sinking velocity of marine snow aggregates. *Biogeosciences*, 7(9), 2613–2624. https://doi.org/10.5194/bg-7-2613-2010

Iversen, M. H., & Ploug, H. (2013). Temperature effects on carbon-specific respiration rate and sinking velocity of diatom aggregates – potential implications for deep ocean export processes. *Biogeosciences*, *10*(6), 4073–4085. https://doi.org/10.5194/BG-10-4073-2013

Iversen, M. H., & Poulsen, L. K. (2007). Coprorhexy, coprophagy, and coprochaly in the copepods Calanus helgolandicus, Pseudocalanus elongatus, and Oithona similis. *Marine Ecology Progress Series*, *350*, 79-89. https://doi.org/10.3354/meps07095

Iversen, M. H., & Robert, M. L. (2015). Ballasting effects of smectite on aggregate formation and export from a natural plankton community. *Marine Chemistry*, *175*, 18-27. https://doi.org/10.1016/j.marchem.2015.04.009 Jackson, G. A. (1994). Particle trajectories in a rotating cylinder: implications for aggregation incubations. *Deep Sea Research Part I: Oceanographic Research Papers*, *41*(3), 429-437. https://doi.org/10.1016/0967-0637(94)90089-2

Jackson, G. A. (2015). Coagulation in a rotating cylinder. *Limnology and Oceanography: Methods*, *13*(4), 194-201. https://doi.org/10.1002/lom3.10018

Jouandet, M. P., Trull, T. W., Guidi, L., Picheral, M., Ebersbach, F., Stemmann, L., & Blain, S. (2011). Optical imaging of mesopelagic particles indicates deep carbon flux beneath a natural iron-fertilized bloom in the Southern Ocean. *Limnology and Oceanography*, *56*(3), 1130–1140. https://doi.org/10.4319/LO.2011.56.3.1130

Kajihara, M. (1971). Settling velocity and porosity of large suspended particle. *Journal of the Oceanographical Society of Japan 1971 27:4*, 27(4), 158–162. https://doi.org/10.1007/BF02109135

Kawana, K., & Tanimoto, T. (1979). Suspended Particles near the Bottom in Osaka Bay*. *Journal of the Oceanographical Society of Japan*, *35*, 75–81.

Kiko, R., Biastoch, A., Brandt, P., Cravatte, S., Hauss, H., Hummels, R., ... & Stemmann, L. (2017). Biological and physical influences on marine snowfall at the equator. *Nature Geoscience*, *10*(11), 852-858. https://doi.org/10.1038/ngeo3042

Kiko, R., Brandt, P., Christiansen, S., Faustmann, J., Kriest, I., Rodrigues, E., ... & Hauss, H. (2020). Zooplankton-mediated fluxes in the eastern tropical North Atlantic. *Frontiers in Marine Science*, *7*, 358. https://doi.org/10.3389/fmars.2020.00358

Kilps, J. R., Logan, B. E., & Alldredge, A. L. (1994). Fractal dimensions of marine snow determined from image analysis of in situ photographs. *Deep Sea Research Part I: Oceanographic Research Papers, 41*(8), 159–1169.

Kiørboe, T. (2001). Formation and fate of marine snow: small-scale processes with large-scale implications. *Scientia marina*, 65(S2), 57-71. https://doi.org/10.3989/scimar.2001.65s257

Kriest, I., & Evans, G. T. (1999). Representing phytoplankton aggregates in biogeochemical models. *Deep Sea Research Part I: Oceanographic Research Papers, 46*(11), 1841-1859. https://doi.org/10.1016/S0967-0637(99)00032-1

Kriest, I., & Oschlies, A. (2008). On the treatment of particulate organic matter sinking in largescale models of marine biogeochemical cycles. *Biogeosciences*, *5*(1), 55–72. https://doi.org/10.5194/bg-5-55-2008

Laurenceau-Cornec, E. C., le Moigne, F. A. C., Gallinari, M., Moriceau, B., Toullec, J., Iversen, M. H., Engel, A., & de La Rocha, C. L. (2020). New guidelines for the application of Stokes' models to the sinking velocity of marine aggregates. *Limnology and Oceanography*, 65(6), 1264–1285. https://doi.org/10.1002/lno.11388

Laurenceau-Cornec, E. C., Trull, T. W., Davies, D. M., de La Rocha, C. L., & Blain, S. (2015). Phytoplankton morphology controls on marine snow sinking velocity. *Marine Ecology Progress Series*, 520, 35–56. https://doi.org/10.3354/meps11116

Leung, S. W., Weber, T., Cram, J. A., & Deutsch, C. (2021). Variable particle size distributions reduce the sensitivity of global export flux to climate change. *Biogeosciences*, *18*(1), 229-250. https://doi.org/10.5194/bg-18-229-2021

Lick, W., Huang, H., & Jepsen, R. (1993). Flocculation of fine-grained sediments due to differential settling. *Journal of Geophysical Research: Oceans*, 98(C6), 10279-10288. https://doi.org/10.1029/93JC00519

Loisel, H., Mériaux, X., Berthon, J. F., & Poteau, A. (2007). Investigation of the optical backscattering to scattering ratio of marine particles in relation to their biogeochemical

composition in the eastern English Channel and southern North Sea. *Limnology and Oceanography*, 52(2), 739-752. https://doi.org/10.4319/lo.2007.52.2.0739

Lombard, F., Boss, E., Waite, A. M., Uitz, J., Stemmann, L., Sosik, H. M., Schulz, J., Romagnan, J. B., Picheral, M., Pearlman, J., Ohman, M. D., Niehoff, B., Möller, K. O., Miloslavich, P., Lara-Lopez, A., Kudela, R. M., Lopes, R. M., Karp-Boss, L., Kiko, R., ... Appeltans, W. (2019). Globally consistent quantitative observations of planktonic ecosystems. *Frontiers in Marine Science*, 6(MAR), 196. https://doi.org/10.3389/fmars.2019.00196 19448007, ja, Downlo

ded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2022GL099563 by University Of Southampton, Wiley Online Library on [31/10/2022]. See the Terms

and Conditions (https://onlinelibrary.wiley.com/term

are governed by the applicable Creat

Maier-Reimer, E., Mikolajewicz, U., & Winguth, A. (1996). Future ocean uptake of CO 2 : interaction between ocean circulation and biology. *Climate Dynamics*, *12*, 711–721.

Mantovanelli, A., & Ridd, P. v. (2006). Devices to measure settling velocities of cohesive sediment aggregates: A review of the in situ technology. *Journal of Sea Research*, *56*(3), 199–226. https://doi.org/10.1016/J.SEARES.2006.05.002

Marsay, C. M., Sanders, R. J., Henson, S. A., Pabortsava, K., Achterberg, E. P., & Lampitt, R. S. (2015). Attenuation of sinking particulate organic carbon flux through the mesopelagic ocean. *Proceedings of the National Academy of Sciences of the United States of America*, *112*(4), 1089–1094. https://doi.org/10.1073/pnas.1415311112

McCave, I. N. (1984). Size spectra and aggregation of suspended particles in the deep ocean. *Deep Sea Research Part A. Oceanographic Research Papers*, *31*(4), 329-352. https://doi.org/10.1016/0198-0149(84)90088-8

McDonnell, A. M. P., & Buesseler, K. O. (2010). Variability in the average sinking velocity of marine particles. *Limnology and Oceanography*, *55*(5), 2085–2096. https://doi.org/10.4319/lo.2010.55.5.2085 McDonnell, A. M., & Buesseler, K. O. (2012). A new method for the estimation of sinking particle fluxes from measurements of the particle size distribution, average sinking velocity, and carbon content. *Limnology and Oceanography: Methods*, *10*(5), 329-346.

Neukermans, G., Loisel, H., Mériaux, X., Astoreca, R., & McKee, D. (2012). In situ variability of mass-specific beam attenuation and backscattering of marine particles with respect to particle size, density, and composition. *Limnology and oceanography*, *57*(1), 124-144. https://doi.org/10.4319/lo.2012.57.1.0124

Niemeyer, D., Kriest, I., & Oschlies, A. (2019). The effect of marine aggregate parameterisations on global biogeochemical model performance. *Biogeosciences Discussions*, 1-29. https://doi.org/10.5194/bg-2019-122

Nowald, N., Fischer, G., Ratmeyer, V., Iversen, M., Reuter, C., & Wefer, G. (2009). In-situ sinking speed measurements of marine snow aggregates acquired with a settling chamber mounted to the Cherokee ROV. *Oceans 2009-Europe*, *2009*, 1–6. https://doi.org/10.1109/OCEANSE.2009.5278186

Nowald, N., Iversen, M. H., Fischer, G., Ratmeyer, V., & Wefer, G. (2015). Time series of insitu particle properties and sediment trap fluxes in the coastal upwelling filament off Cape Blanc, Mauritania. *Progress in Oceanography*, *137*, 1–11. https://doi.org/10.1016/J.POCEAN.2014.12.015

P Chase, R. R. (1979). Settling behavior of natural aquatic particulates'. *Limnology & Oceanography*, 24(3), 417–426.

Parekh, P., Dutkiewicz, S., Follows, M. J., & Ito, T. (2006). Atmospheric carbon dioxide in a less duty world. *Geophysical Research Letters*, *33*(3). https://doi.org/10.1029/2005GL025098

Picheral, M., Catalano, C., Brousseau, D., Claustre, H., Coppola, L., Leymarie, E., Coindat, J., Dias, F., Fevre, S., Guidi, L., Irisson, J. O., Legendre, L., Lombard, F., Mortier, L., Penkerch, C.,

Rogge, A., Schmechtig, C., Thibault, S., Tixier, T., ... Stemmann, L. (2022). The Underwater Vision Profiler 6: an imaging sensor of particle size spectra and plankton, for autonomous and cabled platforms. *Limnology and Oceanography: Methods, 20*(2), 115-129. https://doi.org/10.1002/lom3.10475

Ploug, H., Grossart, H. P., Azam, F., & Jørgensen, B. B. (1999). Photosynthesis, respiration, and carbon turnover in sinking marine snow from surface waters of Southern California Bight: implications for the carbon cycle in the ocean. *Marine Ecology Progress Series*, *179*, 1–11. https://doi.org/doi:10.3354/meps179001

Ploug, H., Iversen, M. H., & Fischer, G. (2008). Ballast, sinking velocity, and apparent diffusivity within marine snow and zooplankton fecal pellets: Implications for substrate turnover by attached bacteria. *Limnology and Oceanography*, *53*(5), 1878–1886. https://doi.org/10.4319/lo.2008.53.5.1878

Ploug, H., Terbrüggen, A., Kaufmann, A., Wolf-Gladrow, D., & Passow, U. (2010). A novel method to measure particle sinking velocity in vitro, and its comparison to three other in vitro methods. *Limnology and Oceanography: Methods*, 8(8), 386–393. https://doi.org/10.4319/LOM.2010.8.386

Ramondenc, S., Madeleine, G., Lombard, F., Santinelli, C., Stemmann, L., Gorsky, G., & Guidi, L. (2016). An initial carbon export assessment in the Mediterranean Sea based on drifting sediment traps and the Underwater Vision Profiler data sets. *Deep Sea Research Part I: Oceanographic Research Papers*, *117*, 107–119. https://doi.org/10.1016/J.DSR.2016.08.015

Riebesell, U. (1992). The formation of large marine snow and its sustained residence in surface waters. *Limnology and Oceanography*, *37*(1), 63-76. https://doi.org/10.4319/lo.1992.37.1.0063

Sarmiento, J. L. & Gruber, N. (2006). *Ocean Biogeochemical Dynamics*. Princeton, Woodstock: Princeton University Press.

Shanks, A. L., & Trent, J. D. (1980). Marine snow: sinking rates and potential role in vertical flux. *Deep-Sea Research*, 27, 137–143.

Small, L. F., Fowler, S. W., & Onlii, M. Y. (1979). Sinking Rates of Natural Copepod Fecal Pellets. *Marine Biology*, *51*, 233–241.

Smayda, T. (1969). Some measurements of the sinking rate of fecal pellets. *Limnology and Oceanography*, *14*(4), 621–625. https://doi.org/10.4319/LO.1969.14.4.0621

Smayda, T. (1970). The suspension and sinking of phytoplankton in the sea. *Oceanography and Marine Biology: An Annual Review*, 8, 353–414. http://ci.nii.ac.jp/naid/10009425107/en/

Smith, S. J., & Friedrichs, C. T. (2015). Image processing methods for in situ estimation of cohesive sediment floc size, settling velocity, and density. *Limnology and Oceanography*, *13*(5), 250–264. https://doi.org/10.1002/lom3.10022

Steinberg, D. K., Van Mooy, B. A., Buesseler, K. O., Boyd, P. W., Kobari, T., & Karl, D. M. (2008). Bacterial vs. zooplankton control of sinking particle flux in the ocean's twilight zone. *Limnology and Oceanography*, *53*(4), 1327-1338. https://doi.org/10.4319/lo.2008.53.4.1327

Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., ... & Winter,
B. (2019). The Canadian earth system model version 5 (CanESM5. 0.3). *Geoscientific Model Development*, *12*(11), 4823-4873. https://doi.org/10.5194/gmd-12-4823-2019

Syvitski, J. P. M., Asprey, K. W., & Leblanc, K. W. G. (1995). In-situ characteristics of particles settling within a deep-water estuary. *Deep Sea Research Part II: Topical Studies in Oceanography*, *42*(1), 223–256. https://doi.org/10.1016/0967-0645(95)00013-G
Takeuchi, M., Doubell, M. J., Jackson, G. A., Yukawa, M., Sagara, Y., & Yamazaki, H. (2019).
Turbulence mediates marine aggregate formation and destruction in the upper ocean. *Scientific Reports*, *9*(1), 1-8. https://doi.org/10.1038/s41598-019-52470-5

Trent, J., Shanks, A. L., & Silver, M. W. (1978). In situ and laboratory measurements on macroscopic aggregates in Monterey Bay, California1. *Limnology and Oceanography*, *23*(4), 626–635.

Twardowski, M. S., Boss, E., Macdonald, J. B., Pegau, W. S., Barnard, A. H., & Zaneveld, J. R. V. (2001). A model for estimating bulk refractive index from the optical backscattering ratio and the implications for understanding particle composition in case I and case II waters. *Journal of Geophysical Research: Oceans, 106*(C7), 14129-14142. https://doi.org/10.1029/2000JC000404

van der Jagt, H., Friese, C., Stuut, J. B. W., Fischer, G., & Iversen, M. H. (2018). The ballasting effect of Saharan dust deposition on aggregate dynamics and carbon export: Aggregation, settling, and scavenging potential of marine snow. *Limnology and Oceanography*, *63*(3), 1386–1394. https://doi.org/10.1002/LNO.10779

Van der Jagt, H., Wiedmann, I., Hildebrandt, N., Niehoff, B., & Iversen, M. H. (2020). Aggregate feeding by the copepods Calanus and Pseudocalanus controls carbon flux attenuation in the Arctic shelf sea during the productive period. *Frontiers in Marine Science*, *7*, 543124. https://doi.org/10.3389/fmars.2020.543124

Volk, T., & Hoffert, M. I. (1985). Ocean carbon pumps: Analysis of relative strengths and efficiencies in ocean-driven atmospheric CO2 changes. *The carbon cycle and atmospheric CO2: natural variations Archean to present, 32*, 99-110.

Xiang, Y., Lam, P. J., Burd, A. B., & Hayes, C. T. (2022). Estimating mass flux from sizefractionated filtered particles: Insights into controls on sinking velocities and mass fluxes in recent U.S. GEOTRACES cruises. *Global Biogeochemical Cycles*, *36*, e2021GB007292. https://doi.org/10.1029/2021GB007292

Yool, A., Palmiéri, J., Jones, C. G., de Mora, L., Kuhlbrodt, T., Popova, E. E., ... & Sellar, A. A. (2021). Evaluating the physical and biogeochemical state of the global ocean component of

UKESM1 in CMIP6 historical simulations. *Geoscientific Model Development, 14*(6), 3437-3472. https://doi.org/10.5194/gmd-14-3437-2021